Author's response to reviews

Title: Epidemiology of frequent attenders: a 3-year historic cohort study comparing attendance, morbidity and prescriptions of one-year and persistent frequent attenders

Authors:

Frans Th.M. Smits (f.smits@gazocentra.nl)
Henk J. Brouwer (h.j.brouwer@amc.uva.nl)
Gerben ter Riet (g.terriet@amc.uva.nl)
Henk H.C. van Weert (h.c.vanweert@amc.uva.nl)

Version: 4 Date: 8 January 2009

Author's response to reviews:

January, 8 2009

Editor of Biomed Medical Central Public Health

Dear editor,

I would like to make some minor changes in the manuscript:

Pag.1 -The correct name is: “Division of Clinical Methods and Public Health”.

-Please change the email address of FTMS in: f.t.smits@amc.uva.nl

Pag.12 -reference 15 is mentioned twice.

Pag.16 -ref 18 is now accepted definitively. The correct reference is:


Selected problems and diseases with ICPC-code:

-chronic respiratory problems:

40,9 R96 astma
49,3 R97 allergic rhinitis; hay fever

With kind regards,

On behalf of all authors,
Frans Smits.